%  This is a Latex template file for articles submitted to MathematicS
%  In Action.  It uses the cedram-msia.cls class file, which is available at 
%          http://msia.cedram.org/?lang=en
%
%  The following template indicates the main features of the class file.
%  In order to avoid mistakes in the handling of meta-data (name, title, etc.),
%  please use all the commands (and only those) indicated in the preamble for
%  the title and authors.
%

\documentclass{cedram-msia}

% Authors may use other style files (e.g. to create figures), as long as
% they do not alter the layout of the article.
\usepackage[all]{xypic}


% Insert here your own symbols, as the following ones:
\newcommand{\bbR}{\mathbb{R}}
\newcommand{\bbC}{\mathbb{C}}
\newcommand{\bbZ}{\mathbb{Z}}


%% Title of the article. 
% The optional argument [] is the short version of the title,
% and the mandatory argument {} the title itself
\title[Sample use of the MSIA class]{How to use the \emph{MathematicS
    in Actions} class file\\ A sample \LaTeX\ source file}


%% Authors, addresses and supports.
% The optional argument is for shortened version appearing in the headings. Please
% distinguish between first, middle and last names with the appropriate commands.
\author{\firstname{Yvon} \lastname{Maday}}
\address{Laboratoire Jacques-Louis Lions\\
Universit\'eerre et Marie Curie\\
Boit courrier 187\\
75252 Paris Cedex 05\\
France}
% No support for the second author
%\thanks{}
\email{maday@ann.jussieu.fr}
%
% Repeat the preceding commands for additional authors, commenting out lines
% which should not appear
\author{\firstname{Denis} \middlename{J.} \lastname{Talay}}
\address{Inria Sophia Antipolis\\
2004 route des Lucioles, BP 93,\\
F-06902 Sophia Antipolis\\
France}
\thanks{The first author is supported by INRIA project OMEGA}
\email{denis.talay@inria.fr}


%% Keywords
\keywords{Example, Applied mathematics, Journal}
  
%% Mathematical classification   (2000)
\subjclass{00X99}


\begin{document}

% Abstract.
\begin{abstract} This document is a short user's guide to the \LaTeX
class for articles in \emph{MathematicS in Actions}.
\end{abstract}

% Use the \maketitle command after the abstract
\maketitle


%% Beginning of text

% Example of section
\section{Introduction, meta-data commands}

This is the beginning of our article.

% Example of subsection
\subsection{Title}

The command for the title is: \verb#\title#.
The \verb#\maketitle# command must be put after the abstract.


\subsection{Citations}

The bibliography must be build using bibtex. A sample of a bibtex file
\texttt{biblio.bib} is available on the MSIA website
%% see the \url command
(\url{http://msia.cedram.org/}.

The references must be put inside a bibtex file constructed from this sample and refered to in the article by using the \verb#\cite# command, which produces for example \cite{KNU} or \cite{LL} (see also the comments in \texttt{biblio.bib}).


\section{Figures}

The article being compiled in pdf, the figures must also be in pdf. The inclusion
of the figure is done using the following commands.

The parameter \emph{xxx}, a real number between $0.0$ and $1.0$, indicates the
width the figure should take in the page.
% One can refer to the figure with \ref{refname}, which gives
% for example: 

Figure \ref{exfig} is an example of figure.

\begin{figure}[tbp]
\includegraphics[width=0.3\linewidth]{figure-msia.pdf}
\caption{Example of figure.}
\label{exfig}
\end{figure}


%To refer to a specific definition, theorem, etc., put \label{labelname} inside
%the corresponding environment and use \ref{labelname} in text to point to this
%definition, theorem, etc.

Here is an example:
\begin{theorem}\label{th:true}
  Most theorems are true.
\end{theorem}

\begin{proof}
  Th. \ref{th:true} is obviously true.
\end{proof}

\begin{example}\label{ex:good}
  This should look like a good example.
\end{example}

\begin{remark}
  Can an example like Ex.~\ref{ex:good} give some insight in
  Th.~\ref{th:true}'s proof?
\end{remark}




% The next command determines the bibliography style. Please do not
% change this.
\bibliographystyle{plain}

%This calls all references from the .bib
\nocite{*}

%  This inserts the bib file
\bibliography{biblio}


\end{document}










